Author’s response to reviews

Title: Left Ventricular Diastolic Dysfunction in Nonhuman Primate Model of Dysmetabolism and Diabetes

Authors:

Haihua Gu (guhaihua@crownbio.com)
Yongqiang Liu (liuyongqiang@crownbio.com)
Shuang Mei (shuang.mei@crownbio.com)
Bingdi Wang (Wangbingdi@crownbio.com)
Guofeng Sun (Sunguofeng@crownbio.com)
Xiaoli Wang (Wangxiaoli@crownbio.com)
Yong-Fu Xiao (xiaoyongfu@crownbio.com)
Michael Staup Staup (mstaup@crownbio.com)
Francine Gregoire (fmgregoire@crownbio.com)
Keefe Chng (keefechn@crownbio.com)
Yixin (Jim) Wang (yxwang@crownbio.com)

Version: 2 Date: 22 Sep 2015

Author’s response to reviews:

Dear Editor,

Thank the reviewers and editor for your constructive comments and suggestions and positive decision on the first revision of our manuscript entitled "Left Ventricular Diastolic Dysfunction in Nonhuman Primate Model of Dysmetabolism and Diabetes". We have made additional changes based on your instruction and therefore, submit the 2nd revision of this manuscript for your consideration to publish in Journal of Cardiovascular Disorders.

Should you have any question, please do not hesitate to contact me.

Regards,
Yi-Xin (Jim) Wang